Author’s response to reviews

Title: Relationship between combination antiretroviral therapy regimens and diabetes mellitus-related comorbidities among HIV patients in Gaborone Botswana

Authors:

jose Gaby Tshikuka (jtshikuka@hotmail.com)
Goabaone Rankgoane-Pono (Goaba2000@yahoo.com)
Mgaywa Gilbert Mjungu Damas Magafu (gmdmgaywa@yahoo.com)
Tiny Masupe (Tiny.masupe@mopipi.ub.bw)
Mooketsi Molefi (mooketsimolefi@gmail.com)
Maurice Nsikungu-Kalukul (mauricensikungu@yahoo.fr)
John Thato Tlhakanelo (john.tlhakanelo@mopipi.ub.bw)
Shimeles G Hamda (shimeles.hamda@mopipi.ub.bw)
Vincent Setlhare (vincent.setlhare@mopipi.ub.bw)

Version: 1 Date: 30 Sep 2017

Author’s response to reviews:

Comment 1 Donne:

Ethical approval to collect data from HIV clinics was sought and obtained from the University of Botswana Review Board and the ethics committee of the Ministry of Health and Wellness, Botswana. Permission to consult clinic record books and systems was also sought and obtained from the clinic management. This was a record based study and no consent to participate was required.

Comment 2: Donne

Comment 3: donne